Author's response to reviews

Title: Placenta increta in a second parous woman causing hemoperitoneum in the 26th week of pregnancy: a case report

Authors:

Gentian Vyshka (gvyshka@yahoo.com)
Nuredin Çapari (nurecapari@gmail.com)
Elmaz Shaqiri (eshaqiri@mjeksialiqjore.gov.al)

Version: 3 Date: 16 June 2010

Author's response to reviews: see over
Placenta increta in a second parous woman causing hemoperitoneum in the 26th week of pregnancy

Elmas Shaqiri M.D. a*, Gentian Vyshka M.D. b, Nuredin Çapari M.D. c

a. Department of Forensic Pathology, Institute of Legal Medicine, Rr. Dibrës 372, Tirana, Albania
   E-mail: eshaqiri@mjeksialigjore.gov.al

b. Biomedical & Experimental Department, Faculty of Medicine, Rr. Dibrës 371, Tirana, Albania.
   (*corresponding author)
   E-mail: gvyshka@yahoo.com

c. District Obstetrical Hospital, Saranda, Albania
   E-mail: nurecapari@gmail.com

*Corresponding author

Gentian Vyshka
Address:
Lagja 1 Rr. Kostaq Cipo Pall 2/19
Tirana-Albania
Tel.: + 355 672907082
E-mail: gvyshka@yahoo.com

The case reported herein is one of the rare uterine ruptures that have been microscopically examined in Albania. The case led to uterine rupture, with hemoperitoneum, and urgent hysterectomy was required.

The prevalence of placenta increta in our country is not yet officially described.

The authors have no conflict of interest.

No funding has been collected for the present study.

There are similar cases clinically described in the literature; yet a full anatomic-pathological discussion is rarely performed.

The present case is a particular one, of interest to the obstetricians and gynecologists.

P.S. This is a re-submission not solicited from the editorial office. We submitted our first version of February, 28th. Meanwhile we've made some changes to the manuscript, including another picture as well.